Author's response to reviews

Title: A study of health effects of long-distance ocean voyages on seamen using a data classification approach

Authors:

Yunmei Lu (lym4321@163.com)
Yanhong Gao (gyh25cn@126.com)
Zhongbo Cao (zbcao66@163.com)
Juan Cui (juancui@csbl.bmb.uga.edu)
Zhennan Dong (Dongzhennan@301hospital.com.cn)
Yaping Tian (tianyp@301hospital.com.cn)
Ying Xu (xyn@bmb.uga.edu)

Version: 6 Date: 28 December 2009

Author's response to reviews: see over
Dear editor,

We would like to submit our manuscript entitled “A study of health effects of long-distance ocean voyages on seamen using a data classification approach” for publication as an article in your Journal.

We would like to state that:

1) All authors concur with the submission.
2) The work has not been published elsewhere, either completely, in part, or in another form.
3) The manuscript has not been submitted to another journal.
4) The manuscript does not contain experiments using animals
5) No financial/commercial conflicts of interests.
6) No papers by the authors that are under consideration or in press with other journals and not yet available online.

Thank your very much for your consideration.

Sincerely Yours,

YUNMEI LU